Author's response to reviews

Title: Acceptance of routine or case-based inquiry for intimate partner violence: a mixed method study

Authors:

Heidi Stöckl (Heidi.Stoeckl@lshtm.ac.uk)
Linda Hertlein (linda.hertlein@med.uni-muenchen.de)
Isabelle Himsl (isabelle.himsl@med.uni-muenchen.de)
Nina Ditsch (nina.ditsch@med.uni-muenchen.de)
Carolin Blume (carolin.blume@med.uni-muenchen.de)
Uwe Hasbargen (uwe.hasbargen@med.uni-muenchen.de)
Klaus Friese (klaus.friese@med.uni-muenchen.de)
Doris Stoeckl (doris.stoeckl@helmholtz-muenchen.de)

Version: 4 Date: 21 February 2013

Author's response to reviews:

Thank you very much for the positive comments about our manuscript!

We asked for the assistance of a fluent English speaking colleague, who corrected the manuscript.

Referee 1:
Discretionary Revisions: I would suggest adding a sentence to the discussion that your hypothesis has been proved by your findings.

We added this sentence in the first paragraph of the discussion section.